Author's response to reviews

**Title:** Changes in medicine course curricula in Brazil encouraged by the Program for the Promotion of Changes in Medical School Curricula (PROMED)

**Authors:**

- Patricia Alves De Souza (p.as@ig.com.br)
- Angelica Maria Bicudo Zeferino (AMBZ@FCM.UNICAMP.BR)
- Marco Aurelio Da Ros (ROS@CCS.UFSC.BR)

**Version:** 6  **Date:** 29 October 2008

**Author's response to reviews:** see over
Dear Natalie

We performed the changes suggested by auditors.

Please note that this paper has been revised by S. B. O'Reilly-Wiese who is a native English speaker (British). She holds a PhD from Imperial College (University of London) in environmental chemistry and works professionally as a translator/reviser of scientific texts.

Regards,

Patrícia Alves de Souza.